Risk spillovers and diversification benefits between crude oil and agricultural commodity futures markets

Walid Mensi^{a,b,d}, Mobeen Ur Rehman^{c,d,1}, Remzi Gök^e, Eray Gemici^f, Xuan Vinh Vo^g

 ^a Department of Finance and Accounting, University of Tunis El Manar and IFGT, Tunisia.
 ^b Department of Economics and Finance, College of Economics and Political Science, Sultan Qaboos University, Muscat, Oman. walidmensil@gmail.com

^c Keele Business School, Keele University, Staffordshire, UK
^d Institute of Business Research, University of Economics Ho Chi Minh City, Vietnam Mobeenrehman@live.com

^e Department of Business Administration, Dicle University, Turkey remzi.gok@dicle.edu.tr ^f Department of Business Administration, Gaziantep University, Turkey gemici@gantep.edu.tr

^g Institute of Business Research and CFVG, University of Economics Ho Chi Minh City, Vietnam <u>vinhvx@ueh.edu.vn</u>

Abstract

This study examines the dependence structure and risk spillovers between crude oil and eight major agricultural futures (wheat, corn, soybean coffee, cotton, lumber, cocoa, and live cattle) markets. It also analyzes the potential conditional diversification benefits using a variety of copula functions and Conditional Value at Risk (CoVaR) measure. The results show significant crisis-sensitive and temporal dependence between oil and agricultural markets. Moreover, crude oil shows a symmetric tail dependence with both wheat, corn, soybeans, and cotton futures, whereas oil exhibits an average dependence with coffee. A strong dependence is observed between oil and cocoa (lumber) during bearish (bullish) market conditions. Oil and Live cattle have a symmetric and bidirectional risk spillovers from oil to agricultural markets. Furthermore, the wheat futures contract appears to be the most dominating and vulnerable asset to oil price shocks, followed by lumber and corn futures, respectively, while the live cattle contracts are the least. Finally, an equally weighted portfolio offers the highest diversification benefits at a 5% expected shortfall.

JEL codes: G14

Keywords: commodity prices; spillovers; diversification benefits; copula. Acknowledgement: This research is partly funded by the University of Economics Ho Chi Minh City, Vietnam.

¹ Corresponding author

Risk spillovers and diversification benefits between crude oil and agricultural commodity futures markets

Abstract

This study examines the dependence structure and risk spillovers between crude oil and eight major agricultural futures (wheat, corn, soybean coffee, cotton, lumber, cocoa, and live cattle) markets. It also analyzes the potential conditional diversification benefits using a variety of copula functions and Conditional Value at Risk (CoVaR) measure. The results show significant crisis-sensitive and temporal dependence between oil and agricultural markets. Moreover, crude oil shows a symmetric tail dependence with both wheat, corn, soybeans, and cotton futures, whereas oil exhibits an average dependence with coffee. A strong dependence is observed between oil and cocoa (lumber) during bearish (bullish) market conditions. Oil and Live cattle have a symmetric dependence during bearish and bullish market conditions. On the other hand, we find asymmetric and bidirectional risk spillovers from oil to agricultural markets. Furthermore, the wheat futures contract appears to be the most dominating and vulnerable asset to oil price shocks, followed by lumber and corn futures, respectively, while the live cattle contracts are the least. Finally, an equally weighted portfolio offers the highest diversification benefits at a 5% expected shortfall.

JEL codes: G14

Keywords: commodity prices; spillovers; diversification benefits; copula

Risk spillovers and diversification benefits between crude oil and agricultural commodity futures markets

1. Introduction

Over the past decade, the global financial system has faced increased uncertainty, resulting in heightened volatility in both crude oil and agricultural commodity markets, with significant implications for financial system stability (Ziadat et al, 2023; Luo and Ji, 2018). The high volatility of crude oil prices can be attributed to various factors, including global demand and supply dynamics, financial crises, and geopolitical conflicts (Alomari et al., 2022; Kang et al., 2017; Mensi et al. 2023). As crude oil serves as a primary energy source for agricultural production (Jiang et al., 2022, Shahzad et al., 2018) rising oil prices impact agricultural commodity prices through higher production costs driven by increased energy requirements, affecting inputs such as fertilizer

and transportation (Natanelov et al., 2011; Silvennoinen and Thorp, 2016; Yip et al., 2020). Therefore, there is a direct link between the energy sector and crop production costs due to the energy intensive nature of agriculture (Koirala et al, 2015; Reboredo, 2012), and this link further extends to impact the pricing of alternative oil substitutes, including biofuels such as Ethanol and Biodiesel, derived from agricultural outputs (Shahzad et al., 2018).

Between September 2003 and July 2008, crude oil prices experienced a significant surge, rising from less than \$30 per barrel to reach a record high of approximately \$147 (Ji et al., 2018; Sun and Shi, 2015). The prices of agricultural products exhibit similar fluctuation patterns to crude oil (Fowowe, 2016; Gong et al., 2023; Rezitis, 2015). For instance, the volatility in oil prices has had a cascading effect on crucial agricultural commodities, such as soybean, maize, and wheat, leading to a reversion of their prices to 2007 levels (Fasanya and Akinbowale, 2019; Nazlioglu and Soytas, 2011). Additionally, during the recent COVID-19 pandemic, commodity prices, including agricultural commodities, experienced notable increases. For instance, rice prices rose by 217%, wheat prices by 136%, corn prices by 125%, and soybean prices by 107% (El Montasser et al, 2023).

The main driving force behind this increase is the rising cost of oil substitutes, such as biofuels, is driving agricultural commodity prices higher (Dahl et al., 2020; Shahzad et al. 2018; Tiwari et al., 2022). Moreover, changes in biofuel prices can also affect fossil fuel prices by affecting demand through the substitution channel, which can lead to bidirectional volatility spillovers across markets, and a better understanding of these relationships can benefit consumers and producers in the long run (Han et al., 2020; Mensi et al., 2014). The substitution relationship between fossil fuels and biofuels is economically reasonable. For example, as a result of the substitution effect between fuel and bio-energy, an increase in biofuel production would exert downward pressure on high oil prices (Su et al., 2019). On the other hand, higher oil prices increase the demand for ethanol, a type of biofuel, leading to higher prices for agricultural products in a mechanism that accepts agricultural products as inputs to energy sources (Paris, 2018; Peñaranda and Micola, 2009). Therefore, producing biofuels to supplement conventional fuels could increase food prices and create sharp increases and volatility in the futures prices of corn, soybeans and crude oil (Chang and Su, 2010).

Notably, the US ethanol policy in 2005 (expanded in 2007 by the Energy Independence and Security Act to include more renewable fuel mixes), envisioning a preference for clean and lowcost ethanol over traditional hydrocarbons, has increased interest in examining the dynamics between energy and agricultural commodity markets. A serious of studies indicate that presence of unidirectional return/volatility spillovers from crude oil to agricultural commodity markets (e.g., Ji et al., 2018; Pal and Mitra, 2020; Zhang and Qu, 2015) while others find evidence that there is both unidirectional and bidirectional return/volatility spillovers from agricultural commodity markets to crude oil markets (e.g., Dahl et al., 2020; Du et al., 2011; Han et al., 2020; Kang et al., 2019; Nazlioglu et al., 2013; Naaem et al., 2022; Shahzad et al., 2018; Tiwari et al., 2022; Yip et al., 2020). This paper extends the understanding of linkages between crude oil and agricultural commodity markets.

Amidst the heightened interdependence of international equity markets brought about by the global financial crisis of 2007-2008, investors have been actively exploring alternative instruments to

diversify their portfolios and mitigate equity risk (Rehman et al. 2023; Makkonen et al., 2021). While the increased volatility spillover between crude oil and agricultural markets complicates risk management for portfolio managers and agricultural producers (Yip et al., 2020), the historically low correlation between energy and agricultural commodities, along with conventional financial assets, has drawn the interest of investors seeking diversification opportunities (Pal and Mitra, 2019). This has led to the recognition of commodity futures as a distinct financial asset class, contributing to effective portfolio diversification (Rehman and Vo, 2020; Kang et al., 2017; Silvennoinen and Thorp, 2016). Particularly, agricultural commodity futures, as an early and crucial component of the futures market, provide a means to hedge or mitigate price risk, minimizing potential losses resulting from adverse price fluctuations (Dai et al., 2022; Liu et al., 2019; Makkonen et al., 2021).

A voluminous literature has considered the returns and/or volatility connectedness between agricultural commodity markets and crude oil (e.g., Dahl et al., 2020; Diebold and Yilmaz, 2017; Hung, 2021; Ji and Fan, 2012; Khalfaoui et al., 2023; Mensi et al., 2014; Nazlioglu et al., 2013; Shahzad et al., 2018; Sun et al., 2021; Tiwari et al., 2022). However, there is no study investigates the pricing dynamics of related markets in terms of tail dependency, risk spillovers and diversification benefits. Therefore, the main motivation of this paper is to fill the gap in the existing literature by examining tail dependency, risk spillovers and diversification benefits between crude oil and agricultural commodity futures markets. In doing so, this study enhances our understanding of the relationship between these markets, and evaluates the potential diversification benefits of including agricultural commodity futures in an oil portfolio using a dynamic conditional diversification benefit approach.

This study makes significant contributions to the existing literature on three key aspects. Firstly, we explore the relationship between WTI crude oil and various agricultural commodity futures (such as wheat, corn, soybeans, coffee, cocoa, cotton, lumber, and live cattle) across different market conditions. Our analysis focuses on different types of copulas, namely the Normal copula, Student-t copula, Clayton copula, rotated Clayton copula, Gumbel copula, rotated Gumbel copula, and Symmetrized Joe-Clayton (SJC) copula. Each copula captures different aspects of dependence, including zero tail dependence, symmetric tail dependence, asymmetric tail dependence.

Secondly, we employ the CoVaR (Conditional Value at Risk) method, as introduced by Adrian and Brunnermeier (2016), to analyze risk spillovers. Unlike traditional Value at Risk (VaR), CoVaR captures asymmetric bidirectional spillovers (downside/upside) between markets, offering a more precise evaluation of risk in uncertain portfolios. CoVaR measures the maximum potential loss that an investor may experience within a specific time horizon and confidence level, considering both long and short positions. This approach is widely used to examine systemic risk and the transmission of failures across financial markets.

Thirdly, we investigate the potential diversification benefits of adding agricultural commodity futures to an oil portfolio using the conditional diversification benefit (CDB) method introduced by Christoffersen et al. (2012, 2017). This approach considers changing correlation and spillover patterns between oil and agricultural commodity futures, allowing us to examine non-linear

dependence and higher-order moments, especially during extreme market events. The CDB measure overcomes the limitations of static models by providing time-varying optimal weights for diversification, thereby accurately estimating the underlying diversification benefits in different market conditions.

The next section provides related studies in current literature. Section 3 describes data and summarizes descriptive statistics. Section 4 introduces the econometric framework. Section 5 presents and discusses the results obtained from the study. Section 6 concludes the paper.

2. Related studies

Following the 2007-2008 global financial crisis (GFC), examining the transmission dynamics between crude oil and agricultural commodities has become a focus of interest for researchers. There is a significant number of studies using various datasets and econometric methods in this field, but the literature gives heterogeneous results. Given the extensive literature on the topic, we confine the literature to only those studies that focus on spillovers in commodity markets. Table 1 provides a brief overview of the main findings and methodologies used in previous studies.

Table 1: Summary of previous studies

<mark>Study</mark>	Data	Method	Commodity	Key findings
<mark>Wang et al.</mark> (2020)	2000- 2019 Daily	MF-DCCA	WTI crude oil, wheat, copper, gold	The study identifies multifractality cross-correlations between crude oil and all agricultural futures, both before and after COVID 19.
<mark>Jiang et al.</mark> (2022)	2004- 2022 Monthly	TVGCT test DY (2012,2014) spillover index	WTI crude oil futures, agricultural futures index, and three Chinese agricultural futures	The findings unveil unidirectional causality and asymmetric, bidirectional volatility spillovers, particularly during financial crises and natural disasters.
Kang et al. (2017)	2002- DI 2016	ECO-GARCH WTI	crude oil, wheat, corn, silver, gold	
Živkov et al. (2020)	2006- 2019 Daily	Component GARCH Quantile regression Modified ICSS algorithm	Brent crude oil, corn, wheat, soybean, and canola.	Their findings indicate that the short-term impact of the oil market has a slightly stronger influence on agricultural commodities compared to its long-term effect.

The study shows that an increased positive connections,

bidirectional spillovers, and gold/silver transmitting information,

while other commodities receive information during crises.

Aitra (2020)	2018	1	at here a some conference of the first state of the
litta (2020)	- 2018 Daily Křef	soybean, ethano	ethanoi, corn, soybeans, and wheat during energy and food crises
	VARI	3EKK WILloit Europe Bren	
soline, heating	2000- The	study uncovers notable spillove	er effects between energy (WTI Mensi et a. GARCH oil,
.014)	2013 VAR-	-DCC- oil, barley, corn	oil, European Brent oil, gasoline, and heating oil) and cerea
	Daily Ga	ARCH sorghum, and whea	, (barley, corn, sorghum, and wheat) commodity markets at
azlioglu	1986- CI	Oil, corn, sugar, wh The study	r find that there is no transmission of risk between oil and
	2011 V Daily IR	soybe agricult	tural markets before the crisis, but volatility spillsovers to
	F	agricultu	ral commodity markets excluding sugar in the post-crisis
			period.
al (2012)			
al. (2013)			
	Vogotak	sta ette ware	identified as the primary drivers of price
ang et al. 2019)	1990- 2017 Baruník and	volatility in cereals, dairy, meat sugar, vegetable oils	both markets, while also revealing bi-directional and s
<u> </u>	lonthly <u>Křehlík</u>	(2018)	asymmetric connectedness across various frequency bands between the two markets
	2018- DY (12)	and crude oil, wheat, sugar,	The
	2020	velet soybean, corn, oats,	
	Daily was	copper	
	Daily Coherence m	odel	
<mark>ung (2021)</mark> oil	Daily coherence m	odel	study indicates a heightened return spillover between crude the COVID-
<mark>łung (2021)</mark> oil	Daily coherence m	odel al commodity markets during t	study indicates a heightened return spillover between crude he COVID- 19 crisis compared to the pre-COVID outbreak period
Hung (2021) oil	Daily coherence m prices and agricultur	al commodity markets during t crude oil and ten major DN	study indicates a heightened return spillover between crude the COVID- 19 crisis compared to the pre-COVID outbreak period Y The study identifies an asymmetric and bidirectional spillover
l <mark>ung (2021)</mark> oil Dahl et al. 2020)	2020 wa Daily coherence m prices and agricultur 1986- 2016 (09,12,14,15) Daily	al commodity markets during t	study indicates a heightened return spillover between crude the COVID- 19 crisis compared to the pre-COVID outbreak period Y The study identifies an asymmetric and bidirectional spillover relationship between crude oil and agricultural commodities
<mark>lung (2021)</mark> oil Dahl et al. 2020)	2020 wa Daily coherence m prices and agricultur 1986- 2016 (09,12,14,15) Daily spillover inde	al commodity markets during t crude oil and ten major DY	study indicates a heightened return spillover between crude the COVID- 19 crisis compared to the pre-COVID outbreak period Y The study identifies an asymmetric and bidirectional spillover relationship between crude oil and agricultural commodities particularly during periods of financial and economic turmoil
Iung (2021) oil Dahl et al. 2020)	2020 wa Daily coherence m prices and agricultur 1986- 2016 (09,12,14,15) Daily spillover inde	al commodity markets during t crude oil and ten major D x commodities	study indicates a heightened return spillover between crude the COVID- 19 crisis compared to the pre-COVID outbreak period Y The study identifies an asymmetric and bidirectional spillover relationship between crude oil and agricultural commodities particularly during periods of financial and economic turmoil
ung (2021) oil ahl et al. 020)	2020 wa Daily coherence m prices and agricultur 1986- 2016 (09,12,14,15) Daily spillover inde	copper odel al commodity markets during t crude oil and ten major D crude oil and ten major D cx commodities fk and lík (2018), <mark>28 commodity futures</mark>	study indicates a heightened return spillover between crude the COVID- 19 crisis compared to the pre-COVID outbreak period Y The study identifies an asymmetric and bidirectional spillover relationship between crude oil and agricultural commodities particularly during periods of financial and economic turmoil The study reveals that an increased intensity of spillover connections among precious metals, industrial metals, energy
ung (2021) oil ahl et al. 020) Gensi et a. 021)	2020 wa Daily coherence m prices and agricultur 1986- 2016 (09,12,14,15) Daily spillover inde 1997- Barum Křeh 2018 and agr	iodel al commodity markets during t crude oil and ten major D crude oil and ten major D cx ik and lík (2018), 28 commodity futures riculture, markets	study indicates a heightened return spillover between crude the COVID- 19 crisis compared to the pre-COVID outbreak period Y The study identifies an asymmetric and bidirectional spillover relationship between crude oil and agricultural commodities particularly during periods of financial and economic turmoil s The study reveals that an increased intensity of spillover connections among precious metals, industrial metals, energy and livestock in the face of economic and political

The energy sector plays a key role in causing significant impacts on other areas, and there are close connections among energy, industrial metals, and precious metals.



Another strand of research that models the dependence between crude oil and agricultural commodities using copula functions has gained significant momentum in recent years. To accurately managing portfolio risk, it is necessary to measure investor dependencies and tails (Rehman et al., 2022). Copula analysis improves the capture of tail dependencies by accounting for deviations from normality in extreme market conditions, providing a more comprehensive approach (Rehman et al., 2023; Ji et al., 2018; Supper et al., 2020). Reboredo (2012), for example, finds weak dependence between world oil and maize, soybean and wheat prices. Koirala et al. (2015) emphasize that a strong positive correlation between energy futures prices and agricultural commodity futures prices. Mensi, Tiwari et al. (2017) report that time-varying asymmetric tail dependence between the pairs of cereals and between oil and cereals across short-term, mediumterm, and long-term investment horizons. Mensi, Hammoudeh et al. (2017) examine the dependence structure and risk spillovers between crude oil prices and major regional developed stock markets and find that there is a tail dependence between crude oil prices and all stock markets for the raw return series. Jiang et al. (2018) conclude that agricultural raw material markets lead the global oil market, which, in turn, leads metal markets, the dependence structure varies over time, and the financial crisis has a notable impact on the interdependencies among these markets. Yahya et al. (2019) show that the connectedness between oil and agricultural products has increased post-2006 across all frequencies, particularly for longer investment horizons. Mokni and Youssef (2019) reveal that a stronger immediate impact of crude oil prices on agricultural commodity prices compared to delayed effects. Kumar et al. (2021) conclude that a strong correlation between oil market crashes and declines in agricultural commodities during crisis periods.

3. Data and summary statistics

Our dataset, spanning from January 1, 2000 to May 5, 2022, includes daily closing prices of oil and eight agricultural futures, i.e. WTI crude oil and cocoa, coffee, corn, cotton, live cattle, lumber, soybean, and wheat. We extracted it from DataStream terminal. The daily return series are calculated as logarithmic first difference of indices and we plot the price and return dynamics of futures in Figures 1 and 2, respectively. We observe that futures prices, see Figure 1, follow different trend before 2013 and a sudden peak observed around 2007 or 2010, coinciding with the 2007-2009 global financial crisis and Eurozone debt crisis, characterizes futures series. In addition, the effect of health crisis in 2020 on evolution of prices becomes evident for all but cotton futures; U.S. oil futures contract prices had dropped in negative territory and back in positive territory soon after while the agricultural assets, except for cocoa, marked a sharp surge with the onset of the COVID-19 pandemic. Evidence of volatility clustering and fat tails in return series, plotted in Figure **2**, prompt the use of nonparametric models.





Figure 1: Price dynamics of WTI oil and Agricultural commodity futures

2010 2013 2016

2019

2022

0 -50

2001 2004

2007









Figure 2: Return dynamics of WTI oil and Agricultural commodity futures

	Wheat	Corn	Soya beans	Coffee	Сосоа	Cotton	Lumber	Live cattle	WTI
Mean	0.0002	0.0002	0.0002	0.0001	0.0001	0.0001	0.0002	0.0001	0.0004
Maximum	0.2391	0.1680	0.0757	0.3154	0.1938	0.0953	0.2696	0.0664	0.3002
Minimum	-0.2467	-0.1619	-0.1674	-0.2626	-0.1928	-0.1040	-0.4006	-0.1595	-0.2822
Std. Dev.	0.0241	0.0185	0.0156	0.0210	0.0176	0.0195	0.0246	0.0112	0.0266
Skewness	-0.0728	-0.1372	-0.7279	0.4651	-0.1842	0.0096	-0.4205	-1.1385	0.2926
Kurtosis	9.3260	4.9377	6.9595	20.6006	13.3852	1.8276	21.7742	13.3416	16.5329
ЈВ	21771.4***	6121.55***	12652.5***	106411.0***	44869.1***	836.7572***	118816.0***	45840.4***	68484.8***
Correlation	0.1346***	0.1721***	0.1824***	0.0903***	0.0446***	0.1631***	0.0545***	0.0581***	-
ADF	-18.8217***	-16.2502***	-17.2142***	-19.4443***	-19.2537***	-17.7918***	-16.4279***	-18.4695***	-16.1425***
PP	-6335.49***	-6120.46***	-6476.01***	-5683.37***	-6249.95***	-6266.45***	-5417.42***	-5527.04***	-5979.94***
ARCH [20]	870.791***	847.237***	816.335***	598.043***	777.239***	1118.60***	86.211***	143.397***	5716.41***
Q [20]	78.972***	43.051***	38.202***	38.442***	37.780***	37.455**	93.282***	66.263***	71.072***

Table 2: Summary statistics of commodity futures price returns

Notes: ***, ** and * represent significance level at 1%, 5% and 10%, respectively.

We delineate the descriptive statistics of each return series in Table 2. Results reveal that all futures contracts post a positive average return, closes to zero during the sample period. As indicated by standard deviations, the crude oil market is more volatile than agricultural commodity markets. The lumber has the highest volatility, whereas the live cattle contract has the lowest volatility among agricultural futures. Coffee, cotton, and WTI have a right-skewed distribution, while the remaining variables exhibit negative skewness. All futures returns are leptokurtic given positive extreme kurtosis values. Neither agricultural nor oil futures contract follow a normal distribution, as suggested by the JB statistics. Oil market is positively and moderately correlates with all agricultural market contracts, with the strongest and least correlations coefficients for soybeans and cocoa, respectively. Both market returns become stationary given the rejection of null hypothesis of ADF and PP tests at the 1% significance level. The results of ARCH and Ljung–Box tests provide evidence of ARCH and serial correlations effects up to 20 lags in all return cases.

4. Empirical methodology

3.1 Time varying copulas

To detect the presence of average and extreme tail dependence between oil and agriculture commodities, we use bivariate copulas. These copulas² are based on Sklar's theorem, which states that the joint distribution function i.e. $F_{XY}(x, y)$ for two continuous random variables X and Y can be expressed in the form of copula functions C(u, v) and the marginal distribution function of F_X and F_Y acting as random variables.

$$F_{XY}(x, y) = \mathcal{C}(u, v) \tag{1}$$

Above equation comprises of $u = F_X(x)$ and $v = F_Y(y)$ which highlights copula as a multivariate function with uniform marginal representing dependence between two random variables. This function is based on $RanF_x * RanF_y$ under continuous margins. We can also use copulas in connecting marginal to multivariate distribution function and later decompose into its respective copula to measure dependence along with its univariate marginal distributions. We can extract joint

probability density of the variables X and Y using copula density $c(u, v) = \frac{\partial^2 C(u, v)}{\partial u \partial v}$ as highlighted below.

(2)

 $f_{XY}(x,y) = c(u,v)f_Y(y)f_X(x)$

 $^{^{2}}$ Cherubini et al. (2004) introduce copulas. Readers are also referred towards Joe (1997) and Nelson (2006) for understanding copulas.

Equation (2) presents $f_Y(y)$ and $f_X(x)$ as marginal densities of variables Y and X, respectively. Therefore, in order to explain joint densities of the two random variables, information about marginal densities and the copulas is crucial. Another feature of copula is that it contains information about the probability of variables experiencing not only average but joint extreme upward or downward (tail) movement. We illustrate equations for upper (right) and lower (left) tail dependence extracted from the copulas as follows.

$$\lambda_U = \lim_{u \to 1} \Pr[X \ge F_X^{-1}(u) | Y \ge F_Y^{-1}(u)] = \lim_{u \to 1} \frac{1 - 2u + C(u, u)}{1 - u}$$
(3)

$$\lambda_{L} = \lim_{u \to 0} \Pr[X \le F_{X}^{-1}(u) | Y \le F_{Y}^{-1}(u)] = \lim_{u \to 0} \frac{\mathcal{C}(u, u)}{u}$$
(4)

In Equation (3) and (4), λ_U , $\lambda_L \in [0,1]$. The lower (upper) tail dependence suggests a non-zero probability for an extreme (small) large value on first series along with the extreme small (large) value for the second series.

In this paper, we use wide range of copulas with different dependence structures under time varying parameters. This combination comprises of symmetric as well as asymmetric copulas. Among symmetric copulas, these include bivariate Frank copula, Plackett copula with tail independence (or zero tail dependence), Student-*t* copula with equal lower and upper tail dependence and the Normal copula. The class of asymmetric copula includes Gumbel copula with upper tail dependence and lower tail independence, Symmetrized Joe Clayton with a special case of symmetric tail dependence, rotated Gumbel copula with upper tail independence and lower tail dependence. Time varying dependence for all the above-mentioned copulas is estimated by allowing dependence parameters to vary across time according to an evolution equation. For student *t* and Gaussian copula, we specify linear dependence parameter ρ_t , which develops following an ARMA (1, q) process³ as follows.

$$\rho_{t} = \wedge \left(\psi_{0} + \psi_{1}\rho_{t-1} + \psi_{2} \frac{1}{q} \sum_{j=1}^{q} \left(\emptyset^{-1}(\mu_{t-j})\right) \left(\emptyset^{-1}(\nu_{t-j})\right)\right)$$
(5)

In Equation (5), $\wedge (x) = (1 + e^{-x})(1 - e^{-x})^{-1}$ presents a modified logistic transformation for keeping ρ_t in (-1,1). The dependence parameters in Equation (5) are presented through ψ_0 , ψ_1 and ψ_2 denoting a constant, an autoregressive term and an average of the product over the last q observations of the transformed variables, respectively. For student-t copula, the parameter dynamics are presented again in Equation (5) however $\emptyset^{-1}(x)$ is replaced by $t_v^{-1}(x)$. Similar to

³ For more information, see Patton (2006)

Gaussian and Student-*t* copula, dynamics of Gumbel and rotated Gumbel copulas follow an ARMA (1, q) as per the following specification.

$$\delta_{t} = \omega + \beta \delta_{t-1} + \alpha \frac{1}{q} \sum_{j=1}^{q} |u_{t-j} - v_{t-j}|$$
(6)

The specification for the tail dependence of Symmetrized Joe Clayton (SJC) copula is as follows.

$$\lambda_{t}^{u} = \Delta \left(\omega_{u} + \beta_{u} \rho_{t-1} + \alpha_{u} \frac{1}{q} \sum_{j=1}^{q} |u_{t-j} - v_{t-j}| \right)$$
(7)

$$\lambda_{t}^{L} = \Delta \left(\omega_{L} + \beta_{L} \rho_{t-1} + \alpha_{L} \frac{1}{q} \sum_{j=1}^{q} |u_{t-j} - v_{t-j}| \right)$$
(8)

In Equations (7) and (8), $\Delta x = (1 + e^{-x})^{-1}$ which presents logistic transformation for keeping λ_t^u and λ_t^L in the range (-1,1).

3.2 Value at risk and conditional value at risk

We quantify risk spillover between oil and agricultural commodity futures contracts utilizing upside and downside value at risk (VaR) and conditional value at risk (CoVaR) based on copula estimates. The VaR quantifies maximum amount of loss that an investor may experience for a specific period and confidence interval by getting into a long (downside) or a short (upside) position. We highlight expression for downside VaR at period t under the confidence interval $1 - \alpha$ as $r(r_t \le VaR_{\alpha,t}) = \alpha$, estimated using marginal models as $VaR_{\alpha,t} = \mu_t + t_{\upsilon,\eta}^{-1}(\alpha)\sigma_t$. The terms μ_t and σ_t represent conditional mean and standard deviation, respectively derived from the dependence model (i.e. ARFIMA-GARCH in our case). Similarly, we present expression of upside VaR using $Pr(r_t \ge VaR_{1-\alpha,t}) = \alpha$ which results in $VaR_{1-\alpha,t} = \mu_t + t_{\upsilon,\eta}^{-1}(1-\alpha)\sigma_t$. Both equations consists of $t_{\upsilon,\eta}^{-1}(\alpha)$ which highlights αth quantile of the student t distribution.

To measure risk spillover from oil market to agricultural commodities, we use CoVaR measure proposed by Adrian and Brunnermeier (2011) which was later generalized by Girardi and Ergün (2013). The CoVaR measure is capable to highlight risk spillover as it increases with the transmission of failure between the associated markets (see Bisias et al. 2012). In our study, the CoVaR of agricultural commodzities (oil) present the sensitivity of the VaR of agricultural commodity(oil) market due to extreme movements in the oil (agricultural commodity) market(s). Suppose r^a and r^o as returns on agricultural and oil market, then the resulting downside CoVaR for agricultural market is appended as follows.

$$Pr(r_t^a \le CoVaR_{\beta,t}^a | r_t^o \le VaR_{\alpha,t}^o) = \beta$$
(9)

In Equation (9), $VaR_{\alpha,t}^o$ provides α th quantile distribution of the oil returns with $Pr(r_t^o \leq VaR_{\alpha,t}^o) = \alpha$ measuring maximum amount of loss that the oil market may undergo for any specific period under the confidence interval of $1 - \alpha$. Similarly, the upside CoVaR for agricultural market results due to extreme upward returns movements in oil market can be specified as

$$Pr(r_t^a \ge CoVaR_{\beta,t}^a | r_t^o \ge VaR_{1-\alpha,t}^o) = \beta$$
⁽¹⁰⁾

In the above equation, $VaR_{1-\alpha,t}^{o}$ measures maximum amount of loss by considering a short position for a period under the confidence interval of $1 - \alpha$. To present CoVaR in equations (9)-(10) in terms of copulas, the resulting conditional probabilities are as follows.

$$C\left(F_{r_t^a}(CoVaR^a_{\beta,t}), F_{r_t^o}(VaR^o_{\alpha,t})\right) = \alpha\beta$$
(11)

$$1 - F_{r_t^a} (CoVaR_{\beta,t}^a) - F_{r_t^o} (VaR_{1-\alpha,t}^o) + C \left(F_{r_t^a} (CoVaR_{\beta,t}^a), F_{r_t^o} (VaR_{1-\alpha,t}^o) \right)$$
$$= (1 - \alpha)(1 - \beta)$$
(12)

The expressions $F_{r_t^o}$ and $F_{r_t^a}$ highlight marginal distribution for oil and agricultural returns, respectively. According to Reboredo and Ugolini (2015), two steps are involved in the estimation of CoVaR. First step involves solving equation (11) or (12) to get value of $F_{r_t^m}(CoVaR_{\beta,t}^m)$ for copula function and the significance level of VaR and CoVaR i.e. α and β , respectively. Second step computes CoVaR as $F_{r_t^m}^{-1}(F_{r_t^m}(CoVaR_{\beta,t}^m))$ with the help of distribution function for agricultural commodities using ARFIMA-FIGARCH marginal model. Furthermore, to add robustness to our results, we apply K-S test proposed by Abadie (2002) for measuring systemic risk comparison between VaR (VaR_a^{\beta,t}) and the CoVaR (CoVaR_a^{\beta,t}) for both agricultural and oil markets. The application of K-S test enables us to measure difference between two quantile functions based on the empirical distribution function, however, does not rely on any underlying distribution function (Li and Wei, 2018).

$$KS_{mn} = \left(\frac{mn}{m+n}\right)^{\frac{1}{2}} \sup_{x} |F_m(x) - G_n(x)|$$
(13)

 $G_n(x)$ and $F_m(x)$ in the above equation denotes distribution functions for VaR and CoVaR, respectively whereas *n* and *m* highlight size of the two samples.

The null hypothesis of no systemic impact between oil and agricultural commodities and the alternative hypothesis can be described below:

$$H_0: CoVaR^a_{\beta,t} = VaR^a_{\beta,t} \tag{14}$$

$$H_1: CoVaR^a_{\beta,t} < VaR^a_{\beta,t} \tag{15}$$

3.3 Conditional diversification benefits (CDB)

Christoffersen et al. (2012) and Christoffersen and Simutin (2017) proposed a test statistic to measure diversification benefits of assets in a portfolio. This test, commonly known as conditional diversification benefit, uses expected shortfall measure to quantify diversification benefits under probability q and at time t as follows.

$$CDB_{t}(\omega_{t},q) = \frac{\omega_{t}ES_{i,t}(q) + (1 - \omega_{t})ES_{g,t}(q) - ES_{p,t}(\omega_{t},q)}{\omega_{t}ES_{i,t}(q) + (1 - \omega_{t})ES_{g,t}(q) - VaR_{t}(q)}$$
(16)

In the above equation, ω_t represent weight of the financial asset *i* at time *t* in portfolio *p* whereas *ES* denotes expected shortfall.

$$ES_{z,t}(q) = -E\left[r_{z,t} | r_{z,t} \le F_{z,t}^{-1}(q)\right]$$
(17)

In the above equation, z = i, g where $F_{z,t}^{-1}(q)$ represents an inverse distribution function for asset z at time t, $\omega_t ES_{i,t}(q) + (1 - \omega_t)ES_{g,t}(q)$ represents an upper bound for the portfolio's expected shortfall, $ES_{p,t}(\omega_{t,q})$. The $VaR_t(q) = -F_{p,t}^{-1}(q)$ gives information about value at risk (VaR) in a portfolio provided by the *qth* quantile of lower bound of expected shortfall. For this reason, the CDB measure provides value in the range [0,1]. The increasing value of CDB implies greater diversification benefits in a portfolio however such diversification benefits depend on the asset's composition and the probability q of the portfolio. For such reasons, we measure CDB under different combinations of assets weight in a portfolio under a passive trading strategy. The weights are kept constant over time at 5% and 50% corresponding to lower tail and median distribution function under Student t distribution is as follows.

$$ES_{z,t}(q) = -\mu_{z,t} + \frac{\sigma_{z,t}}{q} h(H^{-1}(q)) \left[\frac{\nu + H^{-1}(q^2)}{\nu - 1} \right]$$
(18)

Above equation consists of H and h which highlights cumulative distribution function and standard Student density with v degrees of freedom, respectively for which the associated VaR is appended below.

$$VaR_{t}(q) = -\mu_{p,t} - \sigma_{p,t}H^{-1}(\alpha)$$
(19)

5. Empirical results

5.1. Dependence analysis

A preliminary and essential step for the estimation of dependence between oil and agricultural futures return is to define the marginal model and we present the results in Table 3. It should be noted that the results are based on an ARFIMA-FIGARCH model with skewed student-t distributions since it allows long memory in futures returns without violating normality assumption, where the optimal values of p, q, r and m parameters are defined by minimizing the ACI (p=0,1,2 considered). The findings in Panel A demonstrate evidence in favor of persistence in mean for wheat, corn, and live cattle futures contracts, pointing out the existence of long-range dependence in returns and therefore rendering them predictable. We also find that the AR(1) parameter in the conditional mean equation is statistically significant only for wheat, corn, and live cattle, indicating that past information is rapidly impounding into their current futures prices. Looking at the results from the conditional volatility equation in Panel B, we observe that volatility is quite persistent for the most of returns-wheat, corn, soybeans, coffee, cotton, lumber, and live cattle-as indicated by statistically significant GARCH[Beta1] parameter. Such result shows the volatility in agricultural markets depends not only on past volatility, but also on concurrent shocks to returns. In addition, the estimated fractional differencing parameter d (d-FIGARCH) is statistically significant for both market futures returns, indicating the presence of high degree of persistence in volatility, with crude oil futures being the most persistent, followed by wheat, soybeans, and corn, and cocoa being the least persistent among variables. Asymmetric volatility (leverage) effects are quite evident in most of futures, i.e., wheat, soybeans, coffee, cotton, lumber, live cattle, and WTI contracts. The statistically significant coefficients of the tail parameter reveals evidence of leptokurtic behavior for both markets, indicating that agricultural and oil markets significantly deviate downside or upside from current price levels. With the exception of soybeans and coffee, diagnostic test findings in Panel C demonstrate no ARCH effects in residuals in all cases. In addition, we find evidence against the presence of serial correlation in all but lumber cases in the residuals (O[20]) and other than soybeans, coffee, and cocoa in the squared residuals (Q2[20]) of marginal distribution models. Overall, the results suggest no misspecification errors in marginal models for oil and agricultural futures returns.

	Wheat	Corn	Soybeans	Coffee	Cocoa	Cotton	Lumber	Live cattle	WTI
Panel A: Mean Equation									
Cst[M]	0.0003	0.0004	0.0006***	-0.0000	0.0003*	0.0002	-0.0004	0.0006***	0.0010***
	(0.0001)	(0.0003)	(0.0002)	(0.0002)	(0.0002)	(0.0002)	(0.0003)	(0.0002)	(0.0003)
d-Arfima	-0.0973**	0.0737*	0.0352	0	0.0236	0.0069	0.0205	0.0858***	0.0324
	(0.0404	(0.0431)	(0.0221)	(0.022)	(0.0176)	(0.0289)	(0.0163)	(0.0211)	(0.0315)
AR[1]	0.6312***	0.5076***	0.0656	-0.1639	-0.1221	0.218	-0.176	0.8655***	0.3064
	(0.1195)	(0.0988)	(0.2829)	(0.4669)	(0.3879)	(0.1392)	(0.1408)	(0.023)	(0.1898)
MA[1]	-0.5662***	-0.5982***	0.0022	0.2119	0.0861	-0.3131**	0.2390**	-0.9101***	-0.3691*
	(0.1139)	(0.1198)	(0.3011)	(0.4468)	(0.3988)	(0.1545)	(0.1334)	(0.015)	(0.204)
Panel B: Variance Equatio	n								
Cst[V]	-0.5662***	0.1324***	0.0662***	0.3217**	0.6758	0.1505***	0.2009***	0.1308**	0.0656
	(0.1139)	(0.05)	(0.0238)	(0.1324)	(0.4387)	(0.0563)	(0.0559)	(0.0583)	(0.0502)
d-Figarch	0.4881***	0.4277***	0.4627***	0.2633***	0.2055***	0.3723***	0.2737***	0.3381***	0.8674***
-	(0.1228)	(0.0744)	(0.0885)	(0.0472)	(0.0498)	(0.0534)	(0.0295)	(0.0834)	(0.2521)
ARCH[Phil1]	0.3349***	0.3099***	0.3131***	0.4283***	0.1393	0.3922	0.6264***	0.4146***	0.1259
	(0.0715)	(0.0564)	(0.0416)	(0.0999)	(0.3194)	(0.0559)	(0.0643)	(0.1266)	(0.1629)
GARCH[Beta1]	0.7023***	0.6506***	0.7184***	0.6120***	0.2984	0.6841***	0.7686***	0.6202***	0.8849
	(0.0826)	(0.0872)	(0.0692)	(0.1125)	(0.3533)	(0.0717)	(0.0496)	(0.1441)	(0.0968)
Asymmetry	1.4897***	-0.0205	-0.4480***	0.4028***	0.0241	0.0866***	0.1846***	-1.2431***	-0.3541***
Tail	31.467***	3.1867***	4.2492***	5.4378***	5.8651***	4.3866***	8.5650***	14.340***	2.6976***
Panel C: Diagnostics tests									

Table 3: Estimate results of marginal model

LL AIC	14351.4 -4.9185	15643.9 -5.3618	16756.9 -5.7435	15308.4 -5.2467	16151.1 -5.5357	15124.9 -5.1838	14305.3 -4.9027	18829.2 -6.4541	14156.4 -4.8516
ARCH[20]	0.1047	1.0672	1.7844	1.6686	1.5552	0.9178	0.7371	0.5626	1.0665
	[1.0000]	[0.3775]	[0.0170]	[0.0310]	[0.0543]	[0.5640]	[0.7908]	[0.9392]	[0.3783]
Q[20]	20.3533	27.7038	12.1496	24.5521	15.079	11.5929	30.1618	28.8291	12.4423
	[0.3133]	[0.0667]	[0.8394]	[0.1377]	[0.6565]	[0.8675]	[0.0359]	[0.0505]	[0.8236]
Q2[20]	2.0468 [0.9999]	22.5434 [0.2087]	36.1602 [0.0067]	32.4909 [0.0192]	32.7032 [0.0181]	18.4309 [0.4276]	14.441 [0.6999]	9.7474 [0.9398]	21.7972 [0.2411]

Notes: The above table presents estimates of maximum likelihood along with the standard deviations of parameters under marginal distribution of the model. We use combination of different values between "0" and "2" for selecting the p, q, r and m lag values. LL is the Log-likelihood values of marginal distribution models while Q[20] and Q2[20] denote empirical Ljung Box statistics as autocorrelation of commodities returns and squared return series. ARCH [20] highlights autoregressive conditional heteroscedasticity. ***, ** and * indicates the rejection of the null hypothesis at a 1%, 5% and 10% significance level, respectively.









c) Soybeans















h) Live cattle



i) Oil

Figure 3: Volatility estimates of commodity futures.

In the next step, we calculate time-varying copulas between oil and agricultural futures markets. Parameter estimates to model the dependence structure through seven copula functions are given in Table 4. The best-fitted copula model for each asset pair, including the crude oil and one agricultural futures contract, are determined by minimizing AIC value as was the case in Wen et al. (2017), Reboredo (2018), Mensi et al. (2021), etc. The results, reported in Table 4, demonstrate that the Symmetrized JC copula emerges as the best fit for wheat, corn, soybeans, and cotton futures, whereas the Gaussian copula is the best fit for coffee; the Clayton copula for cocoa; the Rotated Gumbel for lumber, and the Student-t copula for live cattle. Our result differs from Mensi et al. (2017), who discover no tail dependence for undecomposed series of oil-wheat and oil-corn pairs but find evidence of asymmetric tail dependence for oil-corn pair in the long-term. More precisely, four out of eight markets are featured concurrently by the existence of lower and upper (symmetric) tail dependence (see Panel F), while the coffee market is described as the symmetric no-tail dependence (see Panel A). Likewise, the lower tail dependence but upper tail independence emerges for cocoa and lumber, whereas the live cattle market exhibits symmetric-with equal lower and upper-tail dependence. Although both the cocoa and lumber futures contracts co-move with oil market during bearish market circumstances, the results suggest a decoupling pattern when markets are bullish. The time varying SJC copula estimates for wheat, soybeans, and cotton reveal that all parameters β_U , β_L , α_U , α_L are statistically and significantly negative. These findings imply the presence of a high level (β_U and β_L) and time-variant persistence (α_U and α_L) in the dependence. Likewise, the conditional upper tail dependence $\dot{\omega}_{\rm U}$ and conditional lower tail dependence $\dot{\omega}_{\rm L}$ for wheat and cotton significantly negative but positive for soybeans. The results show a higher possibility of joint extreme events during both bearish and bullish circumstances for these markets. Conversely, the time-varying estimated parameters (Ψ_1 and Ψ_2) for Clayton and Student-t copulas appear to be statistically significant for cocoa and live cattle and this result confirms that the dependence between oil markets with cocoa and live cattle varies in time. This result confirms Wen et al. (2017), who find a dynamic dependence and thus time-variant diversification benefits between energy stocks and commodity futures.

	Wheat	Corn	Soybeans	Coffee	Cocoa	Cotton	Lumber	Live cattle
Panel A: Gaussian C	opula (no tail d	ependence, t	ail independ	ence)				
 ò	-0.0004***	-0.0019***	0.6229***	0.0007***	0.0014***	0.5379***	0.1752***	0.2415***
	(0.0005)	(0.0005)	(0.0777)	(0.0004)	(0.0010)	(0.0804)	(0.0619)	(0.0533)
a	0.0092***	0.0089***	0 2448***	0.0121***	0.0107***	0.1754***	0.1950***	0 1492***
u	(0.0029)	(0.0025)	(0.0722)	(0.0036)	(0.0046)	(0.0702)	(0.0770)	(0.0575)
ß	(0.002)) 2.0112***	2 0255***	1 5624***	1 0006***	1 072/***	1 5047***	1 2185***	1 0805***
P	(0.0071)	(0.0045)	(0.2088)	(0.0057)	(0.0204)	(0.2072)	-1.2105	-1.9005
AIC	(0.0071)	(0.0045)	(0.2900)	(0.0057)	(0.0204)	172.005	(0.0124)	(0.0097)
Panel B: Clayton Cor	-155.961 pula (lower tail	-210.20 dependence)	-239.000	-135.502	-29.402	-172.095	-32.097	-33.407
Tuner D. Chayton Co	puiu (ioivei uii	acpenaence)						
Ψ_0	0.2738***	0.7747***	1.0317***	0.8769***	0.7381***	0.4128***	0.7826***	0.2463***
	(0.0558)	(0.1520)	(0.1179)	(0.0389)	(0.0290)	(0.0686)	(0.0260)	(1.0900)
Ψ_1	0.9987***	-0.8141***	-0.6151***	-0.4751***	0.3372***	0.6954***	-0.8855***	1.1313
	(0.1388)	(0.4211)	(0.1959)	(0.0878)	(0.0345)	(0.0870)	(0.1310)	(17.3400)
Ψ_2	-0.1205***	-0.5877***	-1.4132***	-1.3344***	-1.6439***	-0.3676***	-1.2869***	-0.1423
	(0.1222)	(0.3864)	(0.3355)	(0.1130)	(0.0568)	(0.1778)	(0.0509)	(15.1837)
AIC	-107.068	-165.614	-217.335	-105.856	-45.053	-179.078	-41.642	-51.416
Panel C: Rotated Cla	yton Copula (u	pper tail dep	endence)					
ò	0.3435***	0.3951***	1.1239***	0.3744**	0.5155***	0.7098***	0.4881***	0.8018***
	(2.2398)	(0.0218)	(0.0813)	(44177.2)	(0.0651)	(0.1688)	(0.1397)	(0.1046)
α	0.8397***	0.7644***	-0.8779***	0.826	0.6007***	-0.5246***	0.5145***	-1.1763***
	(5.5204)	(0.0426)	(0.1536)	(16317.4)	(1.3893)	(0.5269)	(0.5150)	(0.1524)
β	-0.2543***	-0.3884***	-1.6563***	-0.4115***	-1.1442***	-0.6425***	-0.8515***	-1.7914***
	(1.6657)	(0.0286)	(0.3165)	(7419.30)	(0.7190)	(0.3644)	(0.1495)	(0.2293)
AIC	-117.689	-154.423	-183.57	-83.135	-12.817	-130.011	-30.15	-22.063
Panel D: Gumbel Co	pula (upper tail	dependence	e)					
 Ò.,	-1 0651***	-0.5769	1 8081***	-0 7524***	-0 2432***	-0 6272***	-0 6019***	0.0341
ωu	(136.8242)	(10.8581)	(0.3264)	(27 03 27)	(0.3346)	(0.1464)	(0.1521)	(2.4357)
	1.0700***	(10.0001)	0.0705+++	1.0474+++	0.002***	0.0222***	0.0(20***	(2.1007)
αu	1.2/99***	(0.2975)	-0.9705***	1.04/4***	(0.0420)	(0.1001)	(0.1010)	(2.2287***
	(116.1672)	(0.3875)	(0.2790)	(53.1095)	(0.2430)	(0.1021)	(0.1013)	(2.2852)
β_u	-0.0992	-0.3838	-1.22/0***	-0.3232***	-1.1006***	-0.2645***	-0.6641***	-0.1686***
	(13.7639)	(71.2054)	(0.2723)	(7.6773)	(0.3422)	(0.1151)	(0.1753)	(0.2175)
AIC	-140.819	-180.108	-220.765	-103.483	-20.125	-173.158	-46.462	-47.485
Panel E: Rotated Gu	mbel Copula (lo	wer tail dep	endence)					
ώL	-1.0707***	-0.7073***	1.6701***	-0.6490***	-0.0127**	-0.6096***	0.5114***	-1.1894***
	(0.2011)	(0.6962)	(0.2974)	(0.0052)	(0.3913)	(0.1150)	(2.3554)	(11.4022)
αı.	1 2846***	1.0018***	-0.8614***	0.9730***	0 5304	0.9198***	-0 0304	1.3937***
	(0.1627)	(0.4358)	(0.2511)	(0.0015)	(0.3015)	(0.0793)	(2.0342)	(9.7239)

Table 4: Results of time varying copulas between WTI oil and agricultural commodity futures returns

β_L	-0.1000***	-0.2517***	-1.1351***	-0.3860***	-1.1493	-0.2640***	-0.8154	-0.1453**
	(0.0841)	(0.9409)	(0.2241)	(0.0421)	(0.3280)	(0.0948)	(0.7462)	(5.2767)
AIC	-134.341	-193.49	-249.907	-123.834	-44.885	-211.553	-54.467	-72.157

ὼ _U	-1.9607***	3.0752	2.1672***	-1.8850***	-17.6777***	-0.6338***	-16.4722	-17.9473**
	(1.0875)	(0.0000)	(1.6400)	(2.5125)	(169.4849)	(1.4907)	(59.8665)	(592.9723)
<mark>βυ</mark>	-5.7111***	-24.9990***	-18.7030***	-11.1784***	-0.7289	-12.2094***	-2.4543	-1.6729
	(3.7725)	(0.0000)	(8.5105)	(8.2149)	(57.5066)	(7.5179)	(30.3869)	(161.1151)
$a_{\rm U}$	6.3283***	-4.8725	-6.0066***	-3.3676***	0.0168	3.8596***	-0.0118***	-0.0049
	(3.6769)	(0.0000)	(3.2444)	(33.6485)	(1.6359)	(2.2254)	(1.0019)	(1.1288)
$\dot{\omega}_L$	-2.8264***	-3.0959	0.4443***	0.0657***	0.0168***	-0.7068***	2.2315	-0.5886
	(1.7852)	(0.0000)	(1.0399)	(1.2709)	(1.1759)	(1.3991)	(2.2012)	(1.5337)
<mark>βι</mark>	-4.4374***	-1.2736	-9.0260***	-11.6438***	-24.9999***	-7.7976***	-24.9926	-13.49
	(5.8212)	(0.0000)	(3.9458)	(4.0734)	(5.2122)	(4.5663)	(11.0955)	(7.0381)
aL	11.1837***	10.9586	-6.4895***	-0.6151***	-3.8245***	2.4789***	-6.9475	4.4664
	(12.6654)	(0.0000)	(4.2766)	(5.5010)	(1.9079)	(3.5243)	(6.6655)	(2.8559)
AIC	-156.763	-222.129	-273.329	-119.793	-35.328	-224.655	-42.666	-62.292
Panel G: Student-t	Copula (symmet	ric tail depen	dence)					
Ψ_0	-0.0002***	-0.0017***	-0.0013***	0.0007***	0.1980***	-0.0020***	0.0709***	0.0322***
	(0.0005)	(0.0008)	(0.0051)	(0.0003)	(0.0634)	(0.0016)	(0.0541)	(0.0637)
Ψ_1	0.0063***	0.0048***	0.0056***	0.0067***	0.0570***	0.0036***	0.0394***	0.0164***
	(0.0023)	(0.0019)	(0.0188)	(0.0025)	(0.0589)	(0.0027)	(0.0287)	(0.0243)
Ψ_2	2.0074***	2.0229***	2.0228***	2.0026***	-1.7510***	2.0266***	0.7884***	1.4898***
	(0.0078)	(0.0086)	(0.0401)	(0.0050)	(0.5269)	(0.0155)	(0.8667)	(0.9973)
υ	5.0000***	4.9954***	4.9947***	5.0000***	5.0000***	4.9926***	5.0000***	5.0000***
	(0.2856)	(0.2465)	(0.2497)	(0.2745)	(0.3352)	(0.2345)	(0.2450)	(0.3300)

Panel F: Symmetrized Joe-Clayton Copula (asymmetric tail dependence, upper and lower tail dependence)

Notes: ***, ** and * indicates the rejection of the null hypothesis at a 1%, 5% and 10% significance level, respectively. The minimum AIC value (in bold), adjusted for small-sample bias, indicates the best fit for copulas and q parameter for the time-varying framework is "8". Values in the first row represent parameter estimates for each copula model and values in the second row and parenthesis are standard errors. U and L stands for upper and lower tail dependence, respectively.

Figure 4 demonstrates the evolving dependence structure under the best-fit copula specifications given in Table 4 and it provides deeper insights into time-varying tail dependence between agricultural commodity futures and oil markets. A visual inspection suggests the level of tail dependence is not constant but time-varying throughout the sample period, intensifying during the major events, such as the 2007-2009 financial crisis; the 2010-2012 Eurozone sovereign debt crisis; the 2014 oil prices shock, and the COVID-19 period, supporting Yahya et al. (2019), who show the financial and economic crisis periods considerably affect the oil-agricultural nexus. The two copula specifications that produce the best fit are the Symmetrized JC copula with three and the Rotated Gumbel with two pairs of agricultural commodities. More precisely, the results provide evidence of symmetric tail dependence for live cattle futures (student-t copula); asymmetric tail

dependence for wheat, corn, and cotton futures (Symmetrized JC copula), and zero tail dependence for coffee futures (Gaussian copula) with oil markets. We may infer from zero dependence on coffee futures that it provides a safe-haven property for investors in the oil market during a downturn, where the effect is more prominent during the global financial crisis. The strengthening of lower dependence demonstrates that cotton investors suffer the largest losses, followed by live cattle, corn and wheat futures during difficult times. Conversely, the Clayton and Rotated Gumbel copulas show lower tail dependence but upper tail independence for soybean, cocoa, and lumber futures with the oil markets, indicating that oil market returns co-move in extreme negative returns but decouple from the respective futures during bullish market conditions. This suggests a higher loss potential during a downturn and a lower profit diversification potential when markets are bullish for soybean, cocoa, and lumber futures investors combined with the oil market.





Figure 4: Temporal dependence between WTI oil and agricultural commodities.

5.2. Risk spillovers analysis

The calculation of the conditional VaR (CoVaR) enables us to evaluate the impact of upside and downside price spillovers on the agricultural futures markets from the crude oil market and vice versa. To do so, the temporal dynamics of the downside and upside CoVaR and VaR estimates for 0.05 and 0.95 values for both α and β , are calculated and given in Figure 5. As indicated by Reboredo (2018), the price spillover effects, determined by the dependence structure between the oil and agricultural markets, emerge in the case of deviations in CoVaR from VaR values. The wider the departure, the stronger the spillover risk effect that occurs between two markets. We

should note that the red and green solid lines indicate the downside VaR and CoVaR of agricultural or oil market futures, while the blue and gray solid lines reflect the upside VaR and CoVaR of the same.

It is obvious that the extent of spillover effects varies over time for each futures contract and in both directions, with dramatic increases following or during large event occurrences (Reboredo, 2018; Ji et al., 2018; Mensi et al., 2021). The results in Panels A and B indicate that in virtually all situations, downside and upside VaR, as well as conditional VaR values, follow similar upward and downward trends, with minor differences in magnitude in two directions, indicating limited evidence of price spillovers for all but soybeans and cocoa futures with the crude oil market. The visual inspection shows relatively low co-movements between oil and agricultural futures and therefore underscore the presence of tail diversification effects for one market against extreme price fluctuations in other market. Regarding spillovers from the agricultural markets, unconditional downside VaR values for agricultural markets mostly and systematically larger than CoVaR values, and this is true for the unconditional upside VaR compared to CoVaR values, thus reinforcing the findings from Ji et al. (2018). As for spillovers from oil market, however, the results in Panel A suggest that unconditional VaR values are larger than that of CoVaR values for the downward spillovers from the oil market, whereas the opposite is true for the upward spillovers, confirming Meng et al. (2020) for a comparable finding between crude oil and Chinese commodity futures. Overall, the limited evidence of price spillovers and low co-movement between the oil and agricultural markets, driven by a narrower gap between the CoVaR and VaR, have implications for investors in both markets, as they may benefit from the hedging capabilities of oil or agricultural futures during periods of extreme market movements.









Table 5 provides the average upside and downside VaR and CoVaR values for both cases, where the relatively small average values confirms the results depicted in Figure 5. In addition, the average downside and upside VaR values from WTI to agricultural futures range from -0.0453 to 0.0450, but they remain constant for all agricultural futures in the opposite direction for downside and upside risks, with an average of 0.0488 and -0.0471, respectively. The difference between the CoVaR and VaR values suggests a higher spillover effect from agricultural futures to the oil market rather than the other way around, and wheat appears to be by far the most sensitive asset to the upside and downside shocks emanating from the oil market, whereas both downside and upside soybeans risk shocks are the most influential asset on oil market than other agricultural commodities, supporting Ji et al. (2018) for their wheat and soybeans commodities with oil and gas markets.

Table 5: Static upside and downside spillovers between oil and agricultural futures

Upside VaR	Downside VaR	Upside CoVaR	Downside CoVaR
Panel A: Spillovers from WTI oil to agricultural markets			

Wheat	0.0450 (0.0169)	-0.0453 (0.0169)	0.0617 (0.0231)	-0.0273 (0.0102)
Corn	0.0359	-0.0352	0.0449	-0.0400
	(0.0107)	(0.0107)	(0.0134)	(0.0140)
Soybeans	0.0301	-0.0291	0.0206	-0.0551
	(0.0095)	(0.0096)	(0.0064)	(0.0180)
Coffee	0.0376	-0.0376	0.0421	-0.0277
	(0.0100)	(0.0100)	(0.0104)	(0.0069)
Cocoa	0.0347	-0.0340	0.0291	-0.0339
	(0.0085)	(0.0085)	(0.0070)	(0.0087)
Cotton	0.0378	-0.0373	0.0518	-0.0398
	(0.0094)	(0.0093)	(0.0129)	(0.0101)
Lumber	0.0439	-0.0445	0.0415	-0.0444
	(0.0146)	(0.0145)	(0.0132)	(0.0131)
Live cattle	0.0234	-0.0226	0.0258	-0.0130
	(0.0062)	(0.0063)	(0.0070)	(0.0037)
Panel B. Snillovers from agr	icultural to WTI oil markets			
1 unei Di opiniovero from ugri				
Wheat	0.0488	-0.0471	0.0340	-0.0629
Wheat	0.0488 (0.0251)	-0.0471 (0.0254)	0.0340 (0.0168)	-0.0629 (0.0319)
Wheat Corn	0.0488 (0.0251) 0.0488	-0.0471 (0.0254) -0.0471	0.0340 (0.0168) 0.0386	-0.0629 (0.0319) -0.0594
Wheat Corn	0.0488 (0.0251) 0.0488 (0.0251)	-0.0471 (0.0254) -0.0471 (0.0254)	0.0340 (0.0168) 0.0386 (0.0195)	-0.0629 (0.0319) -0.0594 (0.0334)
Wheat Corn Soybeans	0.0488 (0.0251) 0.0488 (0.0251) 0.0488	-0.0471 (0.0254) -0.0471 (0.0254) -0.0471	0.0340 (0.0168) 0.0386 (0.0195) 0.0287	-0.0629 (0.0319) -0.0594 (0.0334) -0.0869
Wheat Corn Soybeans	0.0488 (0.0251) 0.0488 (0.0251) 0.0488 (0.0251)	-0.0471 (0.0254) -0.0471 (0.0254) -0.0471 (0.0254)	0.0340 (0.0168) 0.0386 (0.0195) 0.0287 (0.0140)	-0.0629 (0.0319) -0.0594 (0.0334) -0.0869 (0.0423)
Wheat Corn Soybeans Coffee	0.0488 (0.0251) 0.0488 (0.0251) 0.0488 (0.0251) 0.0488	-0.0471 (0.0254) -0.0471 (0.0254) -0.0471 (0.0254) -0.0471	0.0340 (0.0168) 0.0386 (0.0195) 0.0287 (0.0140) 0.0292	-0.0629 (0.0319) -0.0594 (0.0334) -0.0869 (0.0423) -0.0458
Wheat Corn Soybeans Coffee	0.0488 (0.0251) 0.0488 (0.0251) 0.0488 (0.0251) 0.0488 (0.0251)	-0.0471 (0.0254) -0.0471 (0.0254) -0.0471 (0.0254) -0.0471 (0.0254)	0.0340 (0.0168) 0.0386 (0.0195) 0.0287 (0.0140) 0.0292 (0.0163)	-0.0629 (0.0319) -0.0594 (0.0334) -0.0869 (0.0423) -0.0458 (0.0287)
Wheat Corn Soybeans Coffee Cocoa	0.0488 (0.0251) 0.0488 (0.0251) 0.0488 (0.0251) 0.0488 (0.0251) 0.0488	-0.0471 (0.0254) -0.0471 (0.0254) -0.0471 (0.0254) -0.0471 (0.0254) -0.0471	0.0340 (0.0168) 0.0386 (0.0195) 0.0287 (0.0140) 0.0292 (0.0163) 0.0250	-0.0629 (0.0319) -0.0594 (0.0334) -0.0869 (0.0423) -0.0458 (0.0287) -0.0484
Wheat Corn Soybeans Coffee Cocoa	0.0488 (0.0251) 0.0488 (0.0251) 0.0488 (0.0251) 0.0488 (0.0251) 0.0488 (0.0251)	-0.0471 (0.0254) -0.0471 (0.0254) -0.0471 (0.0254) -0.0471 (0.0254) -0.0471 (0.0254)	0.0340 (0.0168) 0.0386 (0.0195) 0.0287 (0.0140) 0.0292 (0.0163) 0.0250 (0.0124)	-0.0629 (0.0319) -0.0594 (0.0334) -0.0869 (0.0423) -0.0458 (0.0287) -0.0484 (0.0247)
Wheat Corn Soybeans Coffee Cocoa Cotton	0.0488 (0.0251) 0.0488 (0.0251) 0.0488 (0.0251) 0.0488 (0.0251) 0.0488 (0.0251) 0.0488	-0.0471 (0.0254) -0.0471 (0.0254) -0.0471 (0.0254) -0.0471 (0.0254) -0.0471 (0.0254) -0.0471	0.0340 (0.0168) 0.0386 (0.0195) 0.0287 (0.0140) 0.0292 (0.0163) 0.0250 (0.0124) 0.0388	-0.0629 (0.0319) -0.0594 (0.0334) -0.0869 (0.0423) -0.0458 (0.0287) -0.0484 (0.0247) -0.0584
Wheat Corn Soybeans Coffee Cocoa Cotton	0.0488 (0.0251) 0.0488 (0.0251) 0.0488 (0.0251) 0.0488 (0.0251) 0.0488 (0.0251) 0.0488 (0.0251)	-0.0471 (0.0254) -0.0471 (0.0254) -0.0471 (0.0254) -0.0471 (0.0254) -0.0471 (0.0254) -0.0471 (0.0254) -0.0471 (0.0254)	0.0340 (0.0168) 0.0386 (0.0195) 0.0287 (0.0140) 0.0292 (0.0163) 0.0250 (0.0124) 0.0388 (0.0186)	-0.0629 (0.0319) -0.0594 (0.0334) -0.0869 (0.0423) -0.0458 (0.0287) -0.0484 (0.0247) -0.0584 (0.0298)
Wheat Corn Soybeans Coffee Cocoa Cotton Lumber	0.0488 (0.0251) 0.0488 (0.0251) 0.0488 (0.0251) 0.0488 (0.0251) 0.0488 (0.0251) 0.0488 (0.0251) 0.0488	-0.0471 (0.0254) -0.0471 (0.0254) -0.0471 (0.0254) -0.0471 (0.0254) -0.0471 (0.0254) -0.0471 (0.0254) -0.0471 (0.0254) -0.0471	0.0340 (0.0168) 0.0386 (0.0195) 0.0287 (0.0140) 0.0292 (0.0163) 0.0250 (0.0124) 0.0388 (0.0186) 0.0262	-0.0629 (0.0319) -0.0594 (0.0334) -0.0869 (0.0423) -0.0458 (0.0287) -0.0484 (0.0247) -0.0584 (0.0298) -0.0592
Wheat Corn Soybeans Coffee Cocoa Cotton Lumber	0.0488 (0.0251) 0.0488 (0.0251) 0.0488 (0.0251) 0.0488 (0.0251) 0.0488 (0.0251) 0.0488 (0.0251) 0.0488 (0.0251) 0.0488 (0.0251)	-0.0471 (0.0254) -0.0471 (0.0254) -0.0471 (0.0254) -0.0471 (0.0254) -0.0471 (0.0254) -0.0471 (0.0254) -0.0471 (0.0254) -0.0471 (0.0254)	0.0340 (0.0168) 0.0386 (0.0195) 0.0287 (0.0140) 0.0292 (0.0163) 0.0250 (0.0124) 0.0388 (0.0186) 0.0262 (0.0129)	-0.0629 (0.0319) -0.0594 (0.0334) -0.0869 (0.0423) -0.0458 (0.0287) -0.0458 (0.0247) -0.0584 (0.0298) -0.0592 (0.0304)
Wheat Corn Soybeans Coffee Cocoa Cotton Lumber Live cattle	0.0488 (0.0251) 0.0488 (0.0251) 0.0488 (0.0251) 0.0488 (0.0251) 0.0488 (0.0251) 0.0488 (0.0251) 0.0488 (0.0251) 0.0488 (0.0251) 0.0488	-0.0471 (0.0254) -0.0471 (0.0254) -0.0471 (0.0254) -0.0471 (0.0254) -0.0471 (0.0254) -0.0471 (0.0254) -0.0471 (0.0254) -0.0471 (0.0254) -0.0471	0.0340 (0.0168) 0.0386 (0.0195) 0.0287 (0.0140) 0.0292 (0.0163) 0.0250 (0.0124) 0.0388 (0.0186) 0.0262 (0.0129) 0.0269	-0.0629 (0.0319) -0.0594 (0.0334) -0.0869 (0.0423) -0.0458 (0.0287) -0.0458 (0.0287) -0.0484 (0.0247) -0.0584 (0.0298) -0.0592 (0.0304) -0.0398
Wheat Corn Soybeans Coffee Cocoa Cotton Lumber Live cattle	0.0488 (0.0251) 0.0488 (0.0251) 0.0488 (0.0251) 0.0488 (0.0251) 0.0488 (0.0251) 0.0488 (0.0251) 0.0488 (0.0251) 0.0488 (0.0251) 0.0488 (0.0251)	-0.0471 (0.0254) -0.0471 (0.0254) -0.0471 (0.0254) -0.0471 (0.0254) -0.0471 (0.0254) -0.0471 (0.0254) -0.0471 (0.0254) -0.0471 (0.0254) -0.0471 (0.0254)	0.0340 (0.0168) 0.0386 (0.0195) 0.0287 (0.0140) 0.0292 (0.0163) 0.0250 (0.0124) 0.0388 (0.0186) 0.0262 (0.0129) 0.0269 (0.0141)	-0.0629 (0.0319) -0.0594 (0.0334) -0.0869 (0.0423) -0.0458 (0.0287) -0.0484 (0.0287) -0.0484 (0.0247) -0.0584 (0.0298) -0.0592 (0.0304) -0.0398 (0.0230)

Note: The table presents the average VaR and the corresponding CoVaR values. The p-values are presented in brackets.

Afterwards, we utilize the K-S test considering three different tests to find out whether the impact of spillovers under VaR and CoVaR specifications changes or not and report the test statistics alongside with the probability values given in square brackets in Table 6. Given the reported pvalues in the first and second columns are lower than any conventional significance level, we can strongly reject the null hypothesis of equality and thus confirm the existence of statistically difference between unconditional downside/up VaR and conditional downside/up VaR (CoVaR) for both markets and conclude that investors react differently to upside and downside trends. This result confirms those of Kumar et al. (2021), who show the evidence of inequality between the downside/upside VaRs and their respective the downside/upside CoVaR values for all commodity returns and argue that rising uncertainty in oil market will negatively affect the commodity returns. Similarly, the results of Test 3, shown in the third column, show an asymmetry of upside and downside risk spillovers between oil and agricultural markets, indicating that the extent of the upside risk spillover is significantly greater than the downside risk spillover for all agricultural commodity futures. This suggests disentangling downside and upside risks to avoid losses when taking risk and portfolio management decisions. Our findings are partly in line with Shahzad et al. (2018) for the existence of significant and strong asymmetry of upside and downside risk spillovers between oil and strong asymmetry of upside and downside risk spillovers between oil and strong asymmetry of upside and downside risk spillovers between oil and strong asymmetry of upside and downside risk spillovers between oil and strong asymmetry of upside and downside risk spillovers between oil and agricultural markets, including soybeans and wheat.

	Testing Downside Risk Spillover	Testing Upside Risk Spillover	Testing the asymmetry of upside and downside risk
	$H_0: CoVaR_D = VaR_D$	$H_0: CoVaR_U = VaR_U$	$H_0: \frac{CoVaR_D}{VaR_D} = \frac{CoVaR_U}{VaR_U}$
	$H_1: CoVaR_D \neq VaR_D$	H_1 : $CoVaR_U \neq VaR_U$	$H_1: \frac{CoVaR_D}{VaR_D} < \frac{CoVaR_U}{VaR_U}$
Wheat	0.6352	0.4535	0.9955
	[0.0000]	[0.0000]	[0.0000]
Corn	0.1629	0.3316	0.5539
	[0.0000]	[0.0000]	[0.0000]
Soybeans	0.749	0.5558	0.9988
	[0.0000]	[0.0000]	[0.0000]
Coffee	0.5876	0.3032	0.9206
	[0.0000]	[0.0000]	[0.0000]
Cocoa	0.0708	0.4156	0.5509
	[0.0000]	[0.0000]	[0.0000]
Cotton	0.1211	0.5133	0.6918
	[0.0000]	[0.0000]	[0.0000]
Lumber	0.0451	0.1031	0.3034
	[0.0000]	[0.0000]	[0.0000]
Live cattle	0.831	0.2204	0.9818
	[0.0000]	[0.0000]	[0.0000]

 Table 6: Tests of equalities and asymmetries of upside and downside VaR-CoVaR

Note: Table provides the K-S statistic for equality of two cumulative distribution functions. P-values are given in square brackets. Column 1 tests for the equality of VaR and CoVaRs in the down market while Column 2 performs the same test in the up market. Column 3 reports the test results for the asymmetry of upside and downside risks.

5.3. Conditional diversification analysis

Table 7 reports the summary statistics of CDBs for various portfolio weights, based on the suggestions in Christoffersen et al. (2012) and Christoffersen and Simutin (2017) and at the 5% (Panel A) and 50% (Panel B) probability levels for the expected shortfall. The empirical evidence, based on the average values, suggest a reverse V-shaped pattern for all agricultural futures at the 5% and 50% levels, with the exception for live cattle futures offering virtually a same diversification benefits to investors at portfolio compositions of 50% and 80%. In addition, the averaged CDBs values are larger for all cases at the 5% level. The results reveal that the averaged CDBs for wheat, corn, coffee, and cotton are quite similar, whereas the CDBs for the remaining markets differ somewhat in magnitude. The benefits of diversification peak at portfolio compositions of 50% oil and 50% agricultural futures and then declines as the portfolio weights rise. Explicitly, we obtain the least and highest CDBs for portfolio weights of 5% and 50%, respectively, for all agricultural markets, with the exceptions for lumber and live cattle futures. In line with Hanif et al. (2023), lumber attains the highest CDBs for portfolio weights from 0.05 to 0.50, whereas live cattle futures contract acts as the derivative asset offering the highest diversification benefits for the remaining portfolio weights. Lumber (soybeans) futures evidently offers higher (lower) diversification benefits and risk reduction, given a relatively lesser (larger) standard deviation of 0.0246 (0.0602), for an equally weighted portfolio composition when both markets are in bearish circumstances. This result suggests that bivariate portfolios including lumber (soybeans) and oil futures are the most (least) attractive in diversifying the extreme risks of oil markets. However, this is true only for the lumber futures in the normal return quantiles, i.e. at the 50% expected shortfall.

		Wheat	Corn	Soybeans	Coffee	Cocoa	Cotton	Lumber	Live cattle
Panel A: Expected Short	fall at 5%								
Portfolio Weight									
0.05	mean	0.1833	0.1494	0.1242	0.162	0.1548	0.1561	0.2026	0.1113
	sd	[0.0598]	[0.0461]	[0.0417]	[0.0563]	[0.0468]	[0.0438]	[0.0587]	[0.0349]
0.2	mean	0.4599	0.4094	0.3639	0.43	0.4271	0.4215	0.4947	0.3494
	sd	[0.0841]	[0.0788]	[0.0798]	[0.0903]	[0.0800]	[0.0724]	[0.0732]	[0.0726]
0.5	mean	0.6091	0.5899	0.5667	0.602	0.6247	0.594	0.6372	0.5964
	sd	[0.0455]	[0.0497]	[0.0602]	[0.0583]	[0.0414]	[0.0382]	[0.0246]	[0.0528]
0.8	mean	0.4747	0.4931	0.5105	0.4903	0.5368	0.4824	0.4834	0.5981
	sd	[0.0874]	[0.0774]	[0.0791]	[0.0667]	[0.0560]	[0.0631]	[0.0713]	[0.0423]
0.95	mean	0.198	0.217	0.2421	0.2089	0.2438	0.204	0.194	0.3185
	sd	[0.0781]	[0.0748]	[0.0834]	[0.0610]	[0.0683]	[0.0597]	[0.0591]	[0.0680]
Panel B: Expected Shortf	all at 50%								

Table 7: Conditional diversification benefits

Portfolio Weight

0.05	mean	0.0445	0.0349	0.0283	0.0385	0.0363	0.0366	0.0499	0.025
	sd	[0.0173]	[0.0123]	[0.0107]	[0.0160]	[0.0127]	[0.0117]	[0.0179]	[0.0089]
0.20	mean	0.1514	0.1263	0.1067	0.1371	0.1347	0.1314	0.1692	0.1004
	sd	[0.0410]	[0.0343]	[0.0322]	[0.0426]	[0.0366]	[0.0324]	[0.0393]	[0.0288]
0.50	mean	0.2422	0.228	0.2127	0.2382	0.254	0.2302	0.2632	0.2332
	sd	[0.0319]	[0.0324]	[0.0373]	[0.0410]	[0.0283]	[0.0262]	[0.0186]	[0.0356]
0.80	mean	0.16	0.169	0.1793	0.1663	0.1928	0.1616	0.1629	0.2335
		[0.0471]	[0.0436]	[0.0464]	[0.0368]	[0.0349]	[0.0344]	[0.0387]	[0.0298]
0.95	mean	0.0495	0.0549	0.0631	0.0519	0.0629	0.0504	0.0475	0.0884
	sd	[0.0260]	[0.0249]	[0.0292]	[0.0196]	[0.0241]	[0.0189]	[0.0181]	[0.0262]
sd									

Notes: We compute the conditional diversification benefit (CDB) for each bivariate portfolio including oil against agricultural futures contract considering expected shortfall values at the 5% (Panel A) and 50% (Panel B) probability levels. The table reports portfolio weights in first column and CDBs in columns 3–10. In addition, the time-average of the conditional diversification benefit and the standard deviation are given in the first and second row [in square bracket].

6. Conclusion

Crude oil and agricultural markets have been marked by high uncertainty after the financialization of commodities. Furthermore, the geopolitical conflicts and the economic and energy crises have contributed to the large swings of commodity prices. Oil is one of the most important input of agriculture sector. A surge in oil prices leads to an increase in agricultural prices. This linkage requires better understanding and measuring the spillover effects and the changes in the dependence structure between crude oil and major commodity futures. This paper aims to examine the dependence (average and extreme or tail) as well as risk spillovers between crude oil futures and eight major agricultural commodity futures wheat, corn, soybean coffee, cotton, lumber, cocoa, and live cattle. Furthermore, it investigates the potential diversification benefits using a variety of copula functions and Conditional Value at Risk (CoVaR) measure.

We obtain the following results. First, it show evidence of mean and volatility persistence in market returns, where oil market emerges as the most persistence and cocoa is the least. Furthermore, corn, cotton, soybeans, and wheat futures prices have symmetric tail dependence, whereas the coffee market has an average dependence on the oil market. Both cocoa and lumber markets show lower tail dependence, while the live cattle market exhibits concurrent lower and upper tail dependence. They co-move with oil market during bearish market circumstances but decouple when markets are bullish. Time-varying copula results reveal a statistically significant and time-variant dependence for cocoa and live cattle with oil market, confirming Wen et al. (2017). Besides, in line with Yahya et al. (2019), the level of tail dependence is not constant but time varying and intensifies during the major financial. Apart from other markets, the dependence level of coffee and live cattle markets switch sing from positive to negative for shortly periods, coinciding with the general strike in Venezuela, the Iraq war, and the onset of the pandemic, indicating the presence

of safe-haven property for investors in oil market. The results show asymmetric and bidirectional risk spillovers from oil to agricultural markets and demonstrate that the effects of return spillover are more prominent during the COVID-19 pandemic, followed by the GCF and the oil price shock in 2014, with a larger impact during bearish market rather than bullish market conditions. The spillover effect from agricultural markets to oil market is larger than the other way around, indicating that oil market is more susceptible to volatility than agricultural markets during extreme conditions. We find that conditional (CoVaR) and unconditional VaR values exhibit similar upward and downward trends, with minor differences in magnitude, and unconditional downside/upside VaR values mostly and systematically larger than CoVaR values, reinforcing the findings existing literature such as Ji et al. (2018) and Meng et al. (2020). The wheat futures contract, followed by lumber and corn futures, respectively, appears to be the most dominating and vulnerable asset to oil price shocks, while the live cattle contract emerges as the least sensitive. The empirical evidence from the conditional diversification benefits suggest a reverse V-shaped pattern for all agricultural futures at the 5% and 50% expected shortfall levels. The highest benefits are available in equally weighted bivariate portfolios for all cases. Finally, yet importantly, lumber (soybeans) futures evidently offer higher (lower) diversification benefits and risk reduction when both markets are in bearish circumstance.

Our results hold multifaceted implications for agricultural investors and policymakers. From investors' point of view, the asymmetric dependence between oil and agricultural futures markets during extreme market conditions calls for dynamic portfolio management and asset allocation decisions to achieve better diversification benefits. In addition, the presence of dependence linkage of agricultural commodities drives the need for a better understanding of its hedging effects towards uncertainty in oil markets and allows for the precise pricing of futures contracts over various market circumstances. On the other hand, our results will help policy makers to design appropriate policies to minimize the effects of spillover shocks as well as reduce vulnerabilities of the agricultural markets to oil shocks and vice versa. First of all, understanding the major dynamics behind downside and upside risk propagations will enable policymakers to establish effective policies and maintain stability in closely related and mutually dependent markets. Policymakers should first identify whether oil-related risks to agricultural commodity markets are caused by nonmarket or market-specific variables such as supply, demand, or others, and then develop appropriate policies. It is essential to note that the upward and downward dependency observed between the two markets cannot be avoided (Shahzad et al., 2018), given the importance of oil as a global production cost and some agricultural items as a main input for alternative fuels to oil. Therefore, actions to mitigate the potential consequences of events that might disrupt the supplydemand balance of the oil market, which is highly sensitive to geopolitical and economic events, on the country's economy, agricultural production, and investors should be conducted. Excessive speculation on agricultural prices to hedge against the negative consequences of global or regional market variations may result in strong demand and hence excessive volatility in both spot and futures prices. It is proposed that futures markets be more effectively regulated,

agricultural production be increased through subsidies, and non-oil energy sources be developed, or productions be supported through legislation against upside risks; thus, it is possible to reduce the burden on consumers caused by rising inflation due to excessive price fluctuations. On the other hand, for countries that rely heavily on commodity export income, it should be noted that decreases in agricultural commodity prices caused by the oil market may result in a decrease in agricultural production (a fall in producer income), an increase in poverty, and thus a decline in economic growth, as well as long-term social unrest. Investigating the impacts of different uncertainty factors on the risk spillover and dependence linkage between agricultural and oil futures markets over various investment horizons remains on the research agenda.

Reference

- Abadie, A. (2002). Bootstrap tests for distributional treatment effects in instrumental variable models. Journal of the American Statistical Association, 97(457), 284-292.
- Adrian, T., & Brunnermeier, M. K. (2011). CoVaR (No. w17454). National Bureau of Economic Research.
- Alomari, Mohammad, Walid Mensi, Xuan Vinh Vo, and Sang Hoon Kang. 2022. "Extreme Return Spillovers and Connectedness between Crude Oil and Precious Metals Futures Markets: Implications for Portfolio Management." Resources Policy 79:103113. doi: 10.1016/j.resourpol.2022.103113.
- Awartani, Basel, Maghyereh Aktham, and Guermat Cherif. 2016. "The Connectedness between Crude Oil and Financial Markets: Evidence from Implied Volatility Indices." Journal of Commodity Markets 4(1):56–69. doi: 10.1016/j.jcomm.2016.11.002.
- Bisias, D., Flood, M., Lo, A. W., & Valavanis, S. (2012). A survey of systemic risk analytics. Annual Review of Financial Economics, 4(1), 255-296.
- Chang, T. H., & Su, H. M. (2010). The substitutive effect of biofuels on fossil fuels in the lower and higher crude oil price periods. Energy, 35(7), 2807-2813.
- Chen, Zhuo, Bo Yan, and Hanwen Kang. 2022. "Dynamic Correlation between Crude Oil and Agricultural Futures Markets." Review of Development Economics 26(3):1798–1849. doi: 10.1111/rode.12885.
- Cherubini, U., Luciano, E., & Vecchiato, W. (2004). Copula methods in finance. John Wiley & Sons.
- Christoffersen, P., Errunza, V., Jacobs, K., & Langlois, H. (2012). Is the potential for international diversification disappearing? A dynamic copula approach. The Review of Financial Studies, 25(12), 3711-3751.
- Christoffersen, S. E., & Simutin, M. (2017). On the demand for high-beta stocks: Evidence from mutual funds. The Review of Financial Studies, 30(8), 2596-2620.
- Dahl, Roy Endré, Atle Oglend, and Muhammad Yahya. 2020. "Dynamics of Volatility Spillover in Commodity Markets: Linking Crude Oil to Agriculture." Journal of Commodity Markets 20:100111. doi: 10.1016/j.jcomm.2019.100111.

- Dai, Yun-Shi, Ngoc Quang Anh Huynh, Qing-Huan Zheng, and Wei-Xing Zhou. 2022. "Correlation Structure Analysis of the Global Agricultural Futures Market." Research in International Business and Finance 61:101677. doi: 10.1016/j.ribaf.2022.101677.
- Diebold, F. X., Liu, L., & Yilmaz, K. (2017). Commodity connectedness (No. w23685). National Bureau of Economic Research.
- Du, X., Cindy L. Yu, and Dermot J. Hayes. 2011. "Speculation and Volatility Spillover in the Crude Oil and Agricultural Commodity Markets: A Bayesian Analysis." Energy Economics 33(3):497–503. doi: 10.1016/j.eneco.2010.12.015.
- El Montasser, Ghassen, Mohamed Malek Belhoula, and Lanouar Charfeddine. 2023. "Co-Explosivity versus Leading Effects: Evidence from Crude Oil and Agricultural Commodities." Resources Policy 81:103331. doi: 10.1016/j.resourpol.2023.103331.
- Fasanya, Ismail, and Seun Akinbowale. 2019. "Modelling the Return and Volatility Spillovers of Crude Oil and Food Prices in Nigeria." Energy 169:186–205. doi: 10.1016/j.energy.2018.12.011.
- Fowowe, Babajide. 2016. "Do Oil Prices Drive Agricultural Commodity Prices? Evidence from South Africa." Energy 104:149–57. doi: 10.1016/j.energy.2016.03.101.
- Gong, Xu, Yujing Jin, and Tangyong Liu. 2023. "Analyzing Pure Contagion between Crude Oil and Agricultural Futures Markets." Energy 269:126757. doi: 10.1016/j.energy.2023.126757.
- Gong, Xu, Yujing Jin, and Tangyong Liu. 2023a. "Analyzing Pure Contagion between Crude Oil and Agricultural Futures Markets." Energy 269:126757. doi: 10.1016/j.energy.2023.126757.
- Han, Liyan, Jiayu Jin, Lei Wu, and Hongchao Zeng. 2020. "The Volatility Linkage between Energy and Agricultural Futures Markets with External Shocks." International Review of Financial Analysis 68:101317. doi: 10.1016/j.irfa.2019.01.011.
- Hanif, W., Mensi, W., Vo, X. V., BenSaïda, A., Hernandez, J. A., & Kang, S. H. (2023). Dependence and risk management of portfolios of metals and agricultural commodity futures. Resources Policy, 82, 103567.
- Hau, Liya, Huiming Zhu, Rui Huang, and Xiang Ma. 2020. "Heterogeneous Dependence between Crude Oil Price Volatility and China's Agriculture Commodity Futures: Evidence from Quantile-on-Quantile Regression." Energy 213:118781. doi: 10.1016/j.energy.2020.118781.
- Hung, Ngo Thai. 2021. "Oil Prices and Agricultural Commodity Markets: Evidence from Pre and during COVID-19 Outbreak." Resources Policy 73:102236. doi: 10.1016/j.resourpol.2021.102236.
- Iqbal, N., Bouri, E., Grebinevych, O., & Roubaud, D. (2023). Modelling extreme risk spillovers in the commodity markets around crisis periods including COVID19. Annals of Operations Research, 330(1), 305-334.
- Ji, Qiang, and Ying Fan. 2012. "How Does Oil Price Volatility Affect Non-Energy Commodity Markets?" Applied Energy 89(1):273–80. doi: 10.1016/j.apenergy.2011.07.038.
- Ji, Qiang, Elie Bouri, David Roubaud, and Syed Jawad Hussain Shahzad. 2018. "Risk Spillover between Energy and Agricultural Commodity Markets: A Dependence-Switching CoVaR-Copula Model." Energy Economics 75:14–27. doi: 10.1016/j.eneco.2018.08.015.
- Jiang, Wei, Ruijie Gao, and Chao Lu. 2022. "The Analysis of Causality and Risk Spillover between Crude Oil and China's Agricultural Futures." International Journal of Environmental Research and Public Health 19(17):10593. doi: 10.3390/ijerph191710593.

- Jiang, Yonghong, Jiashun Lao, Bin Mo, and He Nie. 2018. "Dynamic Linkages among Global Oil Market, Agricultural Raw Material Markets and Metal Markets: An Application of Wavelet and Copula Approaches." Physica A: Statistical Mechanics and Its Applications 508:265–79. doi: 10.1016/j.physa.2018.05.092.
- Joe, H. (1997). Multivariate models and multivariate dependence concepts. CRC Press.
- Kang, S. H., Tiwari, A. K., Albulescu, C. T., & Yoon, S. M. (2019). Exploring the time-frequency connectedness and network among crude oil and agriculture commodities V1. Energy Economics, 84, 104543.
- Kang, Sang Hoon, Ron McIver, and Seong-Min Yoon. 2017. "Dynamic Spillover Effects among Crude Oil, Precious Metal, and Agricultural Commodity Futures Markets." Energy Economics 62:19–32. doi: 10.1016/j.eneco.2016.12.011.
- Khalfaoui, Rabeh, Umer Shahzad, Mahdi Ghaemi Asl, and Sami Ben Jabeur. 2023. "Investigating the Spillovers between Energy, Food, and Agricultural Commodity Markets: New Insights from the Quantile Coherency Approach." The Quarterly Review of Economics and Finance 88:63–80. doi: 10.1016/j.gref.2022.12.006.
- Koirala, Krishna H., Ashok K. Mishra, Jeremy M. D'Antoni, and Joey E. Mehlhorn. 2015. "Energy Prices and Agricultural Commodity Prices: Testing Correlation Using Copulas Method." Energy 81:430–36. doi: 10.1016/j.energy.2014.12.055.
- Kumar, Satish, Aviral Kumar Tiwari, Ibrahim Dolapo Raheem, and Erik Hille. 2021. "Time-Varying Dependence Structure between Oil and Agricultural Commodity Markets: A Dependence-Switching CoVaR Copula Approach." Resources Policy 72:102049. doi: 10.1016/j.resourpol.2021.102049.
- Li, X., & Wei, Y. (2018). The dependence and risk spillover between crude oil market and China stock market: New evidence from a variational mode decomposition-based copula method. Energy Economics, 74, 565–581.
- Liu, Xiang-dong, Fei Pan, Lin Yuan, and Yu-wang Chen. 2019. "The Dependence Structure between Crude Oil Futures Prices and Chinese Agricultural Commodity Futures Prices: Measurement Based on Markov-Switching GRG Copula." Energy 182:999–1012. doi: 10.1016/j.energy.2019.06.071.
- Luo, Jiawen, and Qiang Ji. 2018. "High-Frequency Volatility Connectedness between the US Crude Oil Market and China's Agricultural Commodity Markets." Energy Economics 76:424–38. doi: 10.1016/j.eneco.2018.10.031.
- Luo, Jiawen, and Qun Zhang. 2021. "Risk Contagions between Global Oil Markets and China's Agricultural Commodity Markets under Structural Breaks." Applied Economics 53(5):628–49. doi: 10.1080/00036846.2020.1808577.
- Makkonen, Adam, Daniel Vallström, Gazi Salah Uddin, Md Lutfur Rahman, and Michel Ferreira Cardia Haddad. 2021. "The Effect of Temperature Anomaly and Macroeconomic Fundamentals on Agricultural Commodity Futures Returns." Energy Economics 100:105377. doi: 10.1016/j.eneco.2021.105377.
- Meng, J., Nie, H., Mo, B., & Jiang, Y. (2020). Risk spillover effects from global crude oil market to China's commodity sectors. Energy, 202, 117208.
- Mensi, W., Hammoudeh, S., Nguyen, D. K., & Yoon, S. M. (2014). Dynamic spillovers among major energy and cereal commodity prices. Energy Economics, 43, 225-243.

- Mensi, W., Rehman, M. U., & Vo, X. V. (2021). Risk spillovers and diversification between oil and nonferrous metals during bear and bull market states. Resources Policy, 72, 102132.
- Mensi, W., Vo, X. V., & Kang, S. H. (2021). Multiscale spillovers, connectedness, and portfolio management among precious and industrial metals, energy, agriculture, and livestock futures. Resources Policy, 74, 102375.
- Mensi, W., Tiwari, A., Bouri, E., Roubaud, D., and Khamis H. Al-Yahyaee. 2017. "The Dependence Structure across Oil, Wheat, and Corn: A Wavelet-Based Copula Approach Using Implied Volatility Indexes." Energy Economics 66:122–39. doi: 10.1016/j.eneco.2017.06.007.
- Mensi, Walid, Mobeen Ur Rehman, Shawkat Hammoudeh, Xuan Vinh Vo, and Won Joong Kim. "How macroeconomic factors drive the linkages between inflation and oil markets in global economies? A multiscale analysis." International Economics 173 (2023): 212-232.
- Mensi, Walid, Shawkat Hammoudeh, Duc Khuong Nguyen, and Seong-Min Yoon. 2014. "Dynamic Spillovers among Major Energy and Cereal Commodity Prices." Energy Economics 43:225–43. doi: 10.1016/j.eneco.2014.03.004.
- Mensi, Walid, Shawkat Hammoudeh, Syed Jawad Hussain Shahzad, and Muhammad Shahbaz. 2017.
 "Modeling Systemic Risk and Dependence Structure between Oil and Stock Markets Using a Variational Mode Decomposition-Based Copula Method." Journal of Banking & Finance 75:258–79. doi: 10.1016/j.jbankfin.2016.11.017.
- Mokni, Khaled, and Manel Youssef. 2020. "Empirical Analysis of the Cross-Interdependence between Crude Oil and Agricultural Commodity Markets." Review of Financial Economics 38(4):635–54. doi: 10.1002/rfe.1096.
- Natanelov, Valeri, Mohammad J. Alam, Andrew M. McKenzie, and Guido Van Huylenbroeck. 2011. "Is There Co-Movement of Agricultural Commodities Futures Prices and Crude Oil?" Energy Policy 39(9):4971–84. doi: 10.1016/j.enpol.2011.06.016.
- Nazlioglu, S., and Ugur Soytas. 2011. "World Oil Prices and Agricultural Commodity Prices: Evidence from an Emerging Market." Energy Economics 33(3):488–96. doi: 10.1016/j.eneco.2010.11.012.
- Nazlioglu, S., Cumhur Erdem, and Ugur Soytas. 2013. "Volatility Spillover between Oil and Agricultural Commodity Markets." Energy Economics 36:658–65. doi: 10.1016/j.eneco.2012.11.009.
- Nelson, R. B. (2006). An Introduction to Copulas, 2nd Ed., Springer- Verlag, New York.
- Nicola, Francesca de, Pierangelo De Pace, and Manuel A. Hernandez. 2016. "Co-Movement of Major Energy, Agricultural, and Food Commodity Price Returns: A Time-Series Assessment." Energy Economics 57:28–41. doi: 10.1016/j.eneco.2016.04.012.
- Pal, D., & Mitra, S. K. 2020. "Time-frequency dynamics of return spillover from crude oil to agricultural commodities." Applied Economics, 52(49), 5426-5445.
- Pal, Debdatta, and Subrata K. Mitra. 2019. "Correlation Dynamics of Crude Oil with Agricultural Commodities: A Comparison between Energy and Food Crops." Economic Modelling 82:453–66. doi: 10.1016/j.econmod.2019.05.017.
- Paris, A. (2018). On the link between oil and agricultural commodity prices: Do biofuels matter?. International Economics, 155, 48-60.
- Patton, A. J. (2006). Modelling asymmetric exchange rate dependence. International economic review, 47(2), 527-556.

- Peñaranda, F., & Micola, A. (2011). On the drivers of commodity co-movement: Evidence from biofuels. Working Paper. University Pompeu Fabra
- Reboredo, J. C. (2018). Green bond and financial markets: Co-movement, diversification and price spillover effects. Energy Economics, 74, 38-50.
- Reboredo, J. C., & Ugolini, A. (2015). Systemic risk in European sovereign debt markets: A CoVaR-copula approach. Journal of International Money and Finance, 51, 214-244.
- Reboredo, Juan C. 2012. "Do Food and Oil Prices Co-Move?" Energy Policy 49:456–67. doi: 10.1016/j.enpol.2012.06.035.
- Rehman, Mobeen Ur, and Xuan Vinh Vo. "Do alternative energy markets provide optimal alternative investment opportunities?" The North American Journal of Economics and Finance 54 (2020): 101271.
- Rehman, Mobeen Ur, Neeraj Nautiyal, Wafa Ghardallou, Xuan Vinh Vo, and Rami Zeitun. "Comovement and spillover among energy markets: A Comparison across different crisis periods." Economic Analysis and Policy (2023).
- Rehman, Mobeen Ur, Paraskevi Katsiampa, Rami Zeitun, and Xuan Vinh Vo. "Conditional dependence structure and risk spillovers between bitcoin and fiat currencies." Emerging Markets Review 55 (2023): 100966.
- Rehman, Mobeen Ur, Peterson Owusu Junior, Nasir Ahmad, and Xuan Vinh Vo. "Time-varying risk analysis for commodity futures." Resources Policy 78 (2022): 102905.
- Rezitis, Anthony N. 2015. "The Relationship between Agricultural Commodity Prices, Crude Oil Prices and US Dollar Exchange Rates: A Panel VAR Approach and Causality Analysis." International Review of Applied Economics 29(3):403–34. doi: 10.1080/02692171.2014.1001325.
- Shahzad, S.J.H., Hernandez, J., Al-Yahyaee, K., and Jammazi. R., 2018. "Asymmetric Risk Spillovers between Oil and Agricultural Commodities." Energy Policy 118:182–98. doi: 10.1016/j.enpol.2018.03.074.
- Silvennoinen, Annastiina, and Susan Thorp. 2016. "Crude Oil and Agricultural Futures: An Analysis of Correlation Dynamics." Journal of Futures Markets 36(6):522–44. doi: 10.1002/fut.21770.
- Su, C. W., Wang, X. Q., Tao, R., & Oana-Ramona, L. (2019). Do oil prices drive agricultural commodity prices? Further evidence in a global bio-energy context. Energy, 172, 691-701.
- Sun, J., and Shi, W., 2015. "Breaks, Trends, and Unit Roots in Spot Prices for Crude Oil and Petroleum Products." Energy Economics 50:169–77. doi: 10.1016/j.eneco.2015.05.001.
- Sun, Yanpeng, Nawazish Mirza, Abdul Qadeer, and Hsin-Pei Hsueh. 2021. "Connectedness between Oil and Agricultural Commodity Prices during Tranquil and Volatile Period. Is Crude Oil a Victim Indeed?" Resources Policy 72:102131. doi: 10.1016/j.resourpol.2021.102131.
- Supper, Hendrik, Felix Irresberger, and Gregor Weiß. 2020. "A Comparison of Tail Dependence Estimators." European Journal of Operational Research 284(2):728–42. doi: 10.1016/j.ejor.2019.12.041.
- Tiwari, A., Abakah, A.E., Adewuyi, A.O., and Lee, CC., 2022. "Quantile Risk Spillovers between Energy and Agricultural Commodity Markets: Evidence from Pre and during COVID-19 Outbreak." Energy Economics 113:106235. doi: 10.1016/j.eneco.2022.106235.
- Umar, Z., Gubareva, M., Naeem, M., & Akhter, A. (2021). Return and volatility transmission between oil price shocks and agricultural commodities. PLoS One, 16(2), e0246886.

- Wang, Jian, Wei Shao, and Junseok Kim. 2020. "Analysis of the Impact of COVID-19 on the Correlations between Crude Oil and Agricultural Futures." Chaos, Solitons & Fractals 136:109896. doi: 10.1016/j.chaos.2020.109896.
- Wen, X., & Nguyen, D. K. (2017). Can investors of Chinese energy stocks benefit from diversification into commodity futures? Economic Modelling, 66, 184-200. doi: 10.1016/j.econmod.2017.06.016.
- Wu, Feng, Zhengfei Guan, and Robert J. Myers. 2011. "Volatility Spillover Effects and Cross Hedging in Corn and Crude Oil Futures." Journal of Futures Markets 31(11):1052–75. doi: 10.1002/fut.20499.
- Yahya, Muhammad, Atle Oglend, and Roy Endré Dahl. 2019. "Temporal and Spectral Dependence between Crude Oil and Agricultural Commodities: A Wavelet-Based Copula Approach." Energy Economics 80:277–96. doi: 10.1016/j.eneco.2019.01.011.
- Yip, Pick Schen, Robert Brooks, Hung Xuan Do, and Duc Khuong Nguyen. 2020. "Dynamic Volatility Spillover Effects between Oil and Agricultural Products." International Review of Financial Analysis 69:101465. doi: 10.1016/j.irfa.2020.101465.
- Ziadat, S.A., Al Rababa'a, A.R.A., Rehman, M. and McMillan, D.G., 2023. Oil price shocks and stock-bond correlation. The North American Journal of Economics and Finance, 68, p.101989.
- Zhang, C., & Qu, X. (2015). The effect of global oil price shocks on China's agricultural commodities. Energy Economics, 51, 354-364.
- Zhu, Huiming, Rong Duan, Cheng Peng, and Xianghua Jia. 2019. "The Heterogeneous Dependence between Global Crude Oil and Chinese Commodity Futures Markets: Evidence from Quantile Regression." Applied Economics 51(28):3031–48. doi: 10.1080/00036846.2018.1564117.
- Živkov, Dejan, Slavica Manić, and Jasmina Đurašković. 2020. "Short and Long-Term Volatility Transmission from Oil to Agricultural Commodities – The Robust Quantile Regression Approach." Borsa Istanbul Review 20:S11–25. doi: 10.1016/j.bir.2020.10.008.

Walid Mensi: Conceptualization, Supervision. Mobeen Ur Rehman: Methodology, Data Curation, Software, Formal analysis. Remzi Gök,: Writing - original draft; administration. Eray Gemici: Formal analysis, Writing - original draft. Xuan Vinh Vo: Supervision, resources, administration

Highlights

- We examine dependence structure and risk spillovers between crude oil and eight major agricultural futures markets.
- Conditional diversification benefits, copula functions and Conditional Value at Risk (CoVaR) measure are used.
- Results show significant crisis-sensitive and temporal dependence between oil and agricultural markets.
- Moreover, crude oil shows a symmetric tail dependence with both wheat, corn, soybeans, and cotton futures
- An equally weighted portfolio offers the highest diversification benefits at a 5% expected shortfall.